Author’s response to reviews

Title: Efficacy of tibolone and raloxifene for the maintenance of skeletal muscle strength, bone mineral density, balance, body composition, cognitive function, mood / depression, anxiety and quality of life / well-being in late postmenopausal women [greater than or equal to] 70 years: Study design of a randomized, double blind, double dummy, placebo-controlled, single-center trial

Authors:

Didy DE Jacobsen (didyjacobsen@hotmail.com)
Monique MM Samson (m.m.samson@umcutrecht.nl)
Yvonne YT van der Schouw (y.t.vanderschouw@umcutrecht.nl)
Diederick DE Grobbee (d.e.grobbee@umcutrecht.nl)
Harald HJJ Verhaar (h.j.j.verhaar@umcutrecht.nl)

Version: 2 Date: 15 November 2007

Author’s response to reviews:

Author’s covering letter for revised submission

Title: Efficacy of tibolone and raloxifene for the maintenance of skeletal muscle strength, bone mineral density, balance, body composition, cognitive function, mood/depression, anxiety and quality of life / well-being in late postmenopausal women [greater than or equal to] 70 years: Study design of a randomized, double-blind, double-dummy, placebo-controlled, single-center trial

Authors:

DE Jacobsen: D.Jacobsen@umcutrecht.nl
MM Samson: M.M.Samson@umcutrecht.nl
YT van der Schouw: Y.T.vanderSchouw@umcutrecht.nl
DE Grobbee: D.E.Grobbee@umcutrecht.nl
HJJ Verhaar: H.J.J.Verhaar@umcutrecht.nl

Version: 2, Date: 15-11-2007

Trials manuscript: T71101S

Dear editor,

In the revised version of the article, a list of abbreviations is added.
Also a legend for figure 1 is provided. The table file is not changed.
The proof of funding and documentation of ethics approval will be sent separately.
Yours sincerely

Ms D.E. Jacobsen, MD
Department of Geriatric Medicine
University Medical Centre Utrecht
PO BOX 85500 (room B05.256)
3508 GA Utrecht, the Netherlands
tel. +31 30 2508280
Email: d.jacobsen@umcutrecht.nl